Author's response to reviews

Title: Study Protocol to Examine the Effects of Spaceflight and a Spaceflight Analog on Neurocognitive Performance: Extent, Longevity, and Neural Bases

Authors:

Vincent Koppelmans (vkoppel@umich.edu)
Burak Erdeniz (berdeniz@umich.edu)
Yiri E De Dios (yiri.e.dedios@nasa.gov)
Scott J Wood (swood@apu.edu)
Patricia A Reuter-Lorenz (parl@umich.edu)
Igor Kofman (igor.kofman-1@nasa.gov)
Jacob J Bloomberg (jacob.j.bloomberg@nasa.gov)
Ajitkumar P Mulavara (ajitkumar.p.mulavara@nasa.gov)
Rachael D Seilder (rseidler@umich.edu)

Version: 4 Date: 5 November 2013

Author's response to reviews:

Dear editor,

By request we made the following changes to our protocol manuscript:

- The index was removed
- The abstract headings Design and Methods were merged into one heading (Methods/Design)

Yours sincerely,

Vincent Koppelmans